Author's response to reviews

Title: The association between farming activities, precipitation, and the risk of acute gastrointestinal illness in rural municipalities of Quebec, Canada: a cross-sectional study

Authors:

Yossi Febriani (yossi.febriani@crchul.ulaval.ca)
Patrick Levallois (patrick.levallois@msp.ulaval.ca)
Suzanne Gingras (suzanne.gingras@inpq.qc.ca)
Pierre Gosselin (pierre.gosselin@inpq.qc.ca)
Shannon E Majowicz (shannon.majowicz@phac-aspc.gc.ca)
Manon D Fleury (Manon_D_Fleury@phac-aspc.gc.ca)

Version: 4 Date: 19 January 2010

Author's response to reviews:

We make little editing changes as requested, only to improve wording.